Author's response to reviews

Title: Etiological study of enteric viruses and the genetic diversity of norovirus, sapovirus, adenovirus, and astrovirus in children with diarrhea in Chongqing, China

Authors:

Zengzhi Ren (renzengzhi@gmail.com)
Yuanmei Kong (kongyuanmei@163.com)
Jun Wang (followbeyond@yahoo.com.cn)
Qianqian Wang (cherish_119@126.com)
Ailong Huang (ahuang1964@yahoo.com.cn)
Hongmei Xu (xuhongmwpx@yahoo.com)

Version: 8 Date: 13 August 2013

Author's response to reviews:

POINT-BY-POINT RESPONSE (MS 8237379818811074)

On submitting the revised manuscript, we want to thank the reviewers and editors for their comments and suggestions, which have helped to shape an improved manuscript. During this revision, we carefully considered all the comments of the editors. Our point-by-point responses to the editors comments are provided below in italic font. The changes are highlighted with underlines in the revised manuscript.

* Please include the Genbank no's of your deposited sequence data in your methods section.
Response: the Genbank no's of deposited sequence data in this article have add in methods section (Page 6, Line 9-12).

* Please include discussion of the limitations of your study in your conclusion section. You should include some of Reviewer 1’s previous concerns regarding time of sampling and length of this here.
Response: The limitation of this study (especially the time of sample collection) has been added in the discussion (Page 12, Line 2-6).

* Please include a competing interest section.
Response: competing interest has been added (Page12, Line 15-16).

* Please include an authors' contributions section.
Response: authors' contribution has been added (Page13, Line 1-6).
